\documentclass[12pt]{article}
\textwidth=6.75in
\textheight=9.25in
\topmargin=-.75in
\oddsidemargin=-10pt
\evensidemargin=-.5in
\parskip=10pt  
\parindent=0pt  

\usepackage{amssymb,latexsym}

 

 

\begin{document} 
 

\noindent
{\bf WEEKLY REPORT 1} %%<------ Change the number as appropriate

{\bf NAME:}  My Name %%<-------- Put your name here


\noindent
{\bf Statement of my problem:}
Please put the problem you are working on here.  You can copy it
word-for-word from the syllabus.

\vskip 10pt
{\bf What I did this week:}

Write down everything you did, what you looked up, progress made, even failed attempts since good can
come from these.

\vskip 10pt
{\bf Questions I have:}

For example,

1. Do you think $\cdots$

2. Where can I find $\cdots$

etc.

\vskip 10pt
{\bf Meeting:}

Can we meet at \underline{\hskip 50pt}?

or

I don't need to meet right now.





\end{document}